
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Wed Feb 28 17:16:29 GMT 2007
Year: 07  Doy: 059
Observer: yasukawa
WEATHER COMMENT: Wed Feb 28 17:16:54 GMT 2007
Clear, SE breeze 3 mph, temp 32 F.  Snow covered Mauna Kea summit and
frost on ground here at MLO.
___end___
**MLSO PROBLEM**: Wed Feb 28 17:18:29 GMT 2007
Evidence of power outage overnight due to thunderstorms over island.
___end___
Wed Feb 28 17:28:49 GMT 2007    CHIP     Start Patrol
Wed Feb 28 17:29:00 GMT 2007    PICS     Start Patrol
PSPT COMMENT: Wed Feb 28 17:31:43 GMT 2007
Starting up.
___end___
SBM COMMENT: Wed Feb 28 17:36:03 GMT 2007
SBM PC was off this morning.  Powering it back up and rebooting.
___end___
Wed Feb 28 17:36:45 GMT 2007    MKIV    Start Patrol
PSPT COMMENT: Wed Feb 28 17:43:10 GMT 2007
Observing
___end___
SBM COMMENT: Wed Feb 28 17:43:20 GMT 2007
Observing.
___end___
Wed Feb 28 18:00:13 GMT 2007    PICS     Flat
**GONG PROBLEM**: Wed Feb 28 18:00:27 GMT 2007
GONG appears down, cannot connect xterm with shelter.  Will go check in a while.
___end___
Wed Feb 28 18:01:48 GMT 2007    CHIP     LSD
Wed Feb 28 18:02:33 GMT 2007    PICS     End Flat
Wed Feb 28 18:02:41 GMT 2007    PICS     ReStart Patrol
Wed Feb 28 18:03:31 GMT 2007    CHIP     End LSD
WEATHER COMMENT: Wed Feb 28 18:03:32 GMT 2007
In deck of clouds blowing off summit.
___end___
Wed Feb 28 18:03:43 GMT 2007    CHIP     BiasLSD
Wed Feb 28 18:04:39 GMT 2007    CHIP     End BiasLSD
Wed Feb 28 18:04:48 GMT 2007    CHIP     Bias
Wed Feb 28 18:05:41 GMT 2007    CHIP     End Bias
Wed Feb 28 18:05:48 GMT 2007    CHIP     ReStart Patrol
Wed Feb 28 18:24:17 GMT 2007    MKIV    End Patrol
Wed Feb 28 18:24:24 GMT 2007    MKIV    Start Cal
GONG COMMENT: Wed Feb 28 18:34:02 GMT 2007
Cycled power to modem/routers.  Xterm connection OK now.



___end___
Wed Feb 28 18:43:12 GMT 2007    MKIV    End Cal
Wed Feb 28 18:43:18 GMT 2007    MKIV    Start Patrol
**CHIP PROBLEM**: Wed Feb 28 18:55:47 GMT 2007
CHIP stalled, Killed and restarted.
___end___
**CHIP PROBLEM**: Wed Feb 28 19:05:02 GMT 2007
Appears MCC needs a reboot.  Stopping PICS and then rebooting MCC.
___end___
WEATHER COMMENT: Wed Feb 28 19:09:11 GMT 2007
In orographic clouds, standing down after restarting PICS and CHIP.
___end___
Wed Feb 28 19:13:06 GMT 2007    MKIV    End Patrol
WEATHER COMMENT: Wed Feb 28 19:13:13 GMT 2007
Closing domes as a precaution.
___end___
Wed Feb 28 19:27:38 GMT 2007    PICS     End Patrol
Wed Feb 28 19:27:41 GMT 2007    CHIP     End Patrol
Wed Feb 28 19:49:41 GMT 2007    MKIV    Start Patrol
WEATHER COMMENT: Wed Feb 28 19:50:09 GMT 2007
Cleared a bit, opened domes and restarting observations.
___end___
Wed Feb 28 19:50:50 GMT 2007    CHIP     Start Patrol
Wed Feb 28 19:51:07 GMT 2007    PICS     Start Patrol
Wed Feb 28 20:22:16 GMT 2007    CHIP     End Patrol
Wed Feb 28 20:22:27 GMT 2007    PICS     End Patrol
WEATHER COMMENT: Wed Feb 28 20:22:28 GMT 2007
In thick clouds again, closing domes and stopping observations.
___end___
Wed Feb 28 20:25:16 GMT 2007    MKIV    End Patrol
WEATHER COMMENT: Wed Feb 28 22:35:19 GMT 2007
In fog and sleet.
___end___
Wed Feb 28 22:59:58 GMT 2007 
        MkIV    
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